Author's response to reviews

Title: Factors associated with paradoxical immune response to antiretroviral therapy in HIV infected patients: a case control study.

Authors:

Janaína A S Casotti (janainacasotti@yahoo.com.br)  
Luciana N Passos (lunaneves@uol.com.br)  
Fabiano J P Oliveira (fabianojpo@yahoo.com.br)  
Crispim Cerutti Jr (fil.cris@terra.com.br)

Version: 5 Date: 1 November 2011

Author's response to reviews:

Dear Sir,

There was a final small, but important, adjustment to be made. In table 4, at the second row, it was necessary to increase the height of the row because the words "previous year" were hidden. Now, I believe the text is ready for publication.

Thank you very much.

Sincerely,

Crispim Cerutti Junior